Author's response to reviews

Title: A t (11; 22) (q24; q12) (EWSR1-FLI1) positive primary vaginal Ewing's sarcoma/Primitive neuroectodermal tumor (PNET) in a young girl: a case report

Authors:

Bharat Rekhi (rekhi.bharat@gmail.com)
Sajid Qureshi (sajidshafique@rediffmail.com)
Ranjan Basak (rb_tiu@yahoo.com)
Sangeeta B. Desai (sangeetabdesai@rediffmail.com)
Seema Medhi (seema.medhi@rediffmail.com)
Purna Kurkure (kurkure@vsnl.com)
Santosh Menon (santosh.menon@rediffmail.com)
Amita Maheshwari (maheshwariamita@yahoo.com)
Nirmala A. Jambhekar (najambhekar@hotmail.com)

Version: 3 Date: 19 January 2010

Author's response to reviews: see over
To,

Professor Dr Michael Kidd

The Editor In Chief, JCMR

Respected Sir,

We thank you and the reviewers for the comments. The additional information has been added across the manuscript. Following are point-wise clarifications:

Q: Please change the main title so it ends simply in 'a case report'

Ans: The title has been suitably modified

Q: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution

Ans: The respective details have been added.

Q: Please include the following sentence in the consent section: 'A copy of the written consent is available for review by the Editor-in-Chief of this journal.'

Ans: This has been added.

Q: Email address: please use the following format for email addresses:

Ans: The email addresses have been formatted.

Q: Please delete the keywords section from the manuscript.

Ans: These have been deleted

Q: Authors' contributions: please ensure this is formatted into a paragraph rather than a list or in bullet points.

Ans: These have been paragraphed

Q: Tables- please remove the visible vertical lines from your tables.

Ans: The vertical grid lines have been removed.
Q: Figure 3: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here:

Ans: Figure 3 has been reformatted.

The manuscript has been proof read. We would be happy to answer any further clarifications, if necessary.

Kind regards,

Dr Bharat Rekhi

Asst Professor/Pathologist,

Tata Memorial Centre, Mumbai, India